Author’s response to reviews

Title: Predicting the risk of pulmonary tuberculosis based on the neutrophil-to-lymphocyte ratio at TB screening in HIV-infected individuals

Authors:
Reiko Miyahara (reiko5556@yahoo.co.jp)
Surachai PIYAWORAWONG (Src_pyw44@hotmail.com)
Vivek NARANBHAI (vnaranbhai@gmail.com)
Prarit PRACHAMAT (Prarit1234@hotmail.com)
Prapimporn KRIENGWATANAPONG (Prapimpornk@gmail.com)
Naho TSUCHIYA (nahot@megabank.tohoku.ac.jp)
Jiraporn WONGYAI (w.jirapohn@gmail.com)
Surasit BUPACHAT (surasit.bupachat@gmail.com)
Norio YAMADA (nyamada@jata.or.jp)
Surin SUMMANAPAN (surinsumanapun@hotmail.com)
Surakameth MAHASIRIMONGKOL (surakameth.m@dmsc.mail.go.th)
Hideki YANAI (yanaih@fukujuji.org)

Version: 2 Date: 20 Jun 2019

Author’s response to reviews:

Dear Dr. Oliver Cocks

Thank you for considering a revised version of our manuscript with Ref. No: INFD-D-18-01931 for publication in BMC infectious diseases. We have addressed the points raised by the editor in this revised version. I need to inform you that we missed to categorize the NL ratio in Table3 and modified it in this revised manuscript. The changes do not affect the conclusion and the discussion. Furthermore, the author information (e-mail address, address of the institution and the telephone number) was changed due to my career change. The e-mail address and telephone number which I wrote in the
submitted manuscript cannot be available from this April. So, I would like to add new institution to my affiliation. The authorship was not changed.

If you require any further information, please let me know.

Yours sincerely,

Reiko Miyahara, MD, PhD
Medical Genomic Science Project
National Center for Global Health and Medicine (NCGM)